Author's response to reviews

**Title:** Parvovirus B19 infection and severe anaemia in Kenyan children: a retrospective case control study

**Authors:**

James W Wildig (jameswildig@optusnet.com.au)
Yvonne Cossart (yvonnecossart@sydney.edu.au)
Norbert Peshu (Npeshu@kemri-wellcome.org)
Nimmo Gicheru (ngicheru@kilifi.kemri-wellcome.org)
James Tuju (jtuju@kilifi.kemri-wellcome.org)
Thomas N Williams (twilliams@kilifi.kemri-wellcome.org)
Charles R Newton (Cnewton@kilifi.kemri-wellcome.org)

**Version:** 7  **Date:** 30 March 2010

**Author's response to reviews:**

Apostrophes added to McNemar's where they were missing. Other very minor corrections.